Author’s response to reviews

Title: A study of the relationship between human infection with avian influenza A (H5N6) and environmental avian influenza viruses in Fujian, China

Authors:

Ping Chen (919958089@qq.com)
JianFeng Xie (xjf417@163.com)
Qi Lin (603089356@qq.com)
Lin Zhao (zhl-xw@tom.com)
Yuanhua Zhang (824479981@qq.com)
Hongbin Chen (80649736@qq.com)
YuWei Weng (wengywfcdc@aliyun.com)
Zheng Huang (Huangz.fj@qq.com)
Kui-Cheng Zheng (kingdadi9909@126.com)

Version: 2 Date: 25 May 2019

Author’s response to reviews:

Dear editor,

Thank you for your letter regarding our manuscript titled

"A study of the relationship between human infection with avian influenza A (H5N6) and environmental avian influenza viruses in Fujian, China ". (Manuscript number: INFD-D-19-00379)

Reviewer reports:

1- Abstract and manuscript deserve one more check for English language.

Response:
We do invite a friend of us who is a native English speaker from USA help polish our article. Due to our friend’s help, the article was edited extensively. And we hope the revised manuscript could be acceptable for you.

2- Acknowledge in the discussion that data on signs and symptoms were not collected.

Response:

We have added them in the revised manuscript. (Discussion section, line 291-295)

Once again, we acknowledge your comments very much, which are valuable in improving the quality of our manuscript. We hope that the updated manuscript will be found suitable for publication in the BMC Infectious Diseases.

Yours Sincerely

Ping Chen